Author's response to reviews

Title: Comparison of Echocardiographic (US) Volumetry with Cardiac Magnetic Resonance (CMR) Imaging in Transfusion Dependent Thalassemia Major (TM)

Authors:

Anastasios Giakoumis (giakanas@otenet.gr)
Vasilis Berdoukas (plomari@hol.gr)
Efstathios Gotsis (sgotsis@hol.gr)
Athanassios Aessopos (aaisopos@cc.uoa.gr)

Version: 3 Date: 10 July 2007

Author's response to reviews:

Formatting changes made:
1. Removed full stop from the end of the title
2. Reformatted affiliations according to recommendations
3. Removed underlinings in any section headings
4. All author names show in References with the exception of 8 and 22 where PubMed had limitations
5. Formated and placed Figure Legends after the References
6. Reformated single composite illustration with parts sub-labeled A-I. Changed also figure referals in the manuscript

Best regards,
The Authors' Team